



25 March 2015 EMA/201512/2015

EU Medicines Agencies Network Strategy to 2020 - Working together to improve health

Submission of comments

Comments from:

Name of organisation or individual

Health Action International

International Society of Drug Bulletins

Medicines in Europe Forum

Please note that these comments and the identity of the sender will be published unless a specific justified objection is received. In your reply please indicate whether you are replying as a citizen, organisation or public authority.

Comments should be sent to the European Medicines Agency electronically and in Word format (not pdf).

Comments should be sent to EUnetworkstrategy@ema.europa.eu and must arrive by 30 June 2015.



General comments

General comment (if any)	Outcome (if applicable)
	<to be="" by="" completed="" ema="" hma="" the=""></to>
 Health Action International (HAI), International Society of Drug Bulletins (ISDB) and Medicines in Europe Forum (MiEF) are pleased to contribute to the public consultation on the EU Medicines Agencies Network Strategy to 2020 (1). In our view, in order to be able to carry out its public health tasks, the EMA needs to: Be weaned off a fee-for-service relationship with pharmaceutical companies through public funding from the European Union; Reconsider its proposal to give systematic scientific advice in exchange of fees which places the Agency in a position of conflict of interest; Focus on evaluating evidence (scientific data) from clinical studies that have been designed to meet health needs, and assess the benefit-harm balance of medicinal products on a comparative basis (therapeutic advance); Improve and enforce its transparency requirements to effectively prevent conflicts of interest and ensure access to regulatory data: pre and post marketing information, clinical data, pharmacovigilance data, as well as a central registry of all data; Encourage the interaction with independent civil society representatives; Prevent expedited marketing authorisations such as adaptive pathways from becoming the rule rather than the exception, if no genuine unmet medical need is at stake, so as to prevent unnecessary exposure to avoidable harm. 	
The independence of the European Medicines Agency is paramount	
In order to understand how EMA's priorities and functioning have evolved, one should be aware that the Agency is very heavily funded by pharmaceutical	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
companies. Industry funding has progressively increased since 1995 when the EMA was established. In 2015, the collection of pharmaceutical companies' fees will amount to more than 83% of the Agency's overall budget (2). In contrast, the fees collected by the US Food and Drug Administration (FDA) from drug companies submitting applications for marketing authorisations for human medicines and/or biological products represent about 60 percent of the FDA's overall budget (3). To guarantee the EMA's independence, and prevent difficulties in sustainability due to fewer applications and subsequent fluctuations in fee revenues, any direct financial relationship between the Agency and industry should be avoided. This could be achieved by restructuring EMA's funding so that fees would make up but a small proportion of its overall budget. Robust policies on conflicts of interest must be in place to safeguard public health	
The independence of the regulatory process is crucial to ensure that public health is not supplanted by private interests. To guarantee independence, medicines agencies and national competent authorities must have in place robust policies of conflicts of interest, for its management board, staff and experts. In this regard, we regret EMA's recent decision to weaken its policy on conflicts of interest for experts. There is no rationale behind the Agency's decision to decrease cool-off periods and to maintain an arbitrary classification system that allows unjustified situations whereby experts with conflicts are permitted to engage in the EMA's policies and their decision-making. Since the EMA is considered a benchmark to many national drug regulatory agencies, we urge them to reconsider and to reverse its policy (4).	
Concrete measures by competent authorities and medicines agencies to increase expertise include:	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
Reinforcing the number and skills of experts who are independent from	
pharmaceutical companies	
2) Significantly reinforcing agencies' in-house expertise.	
3) Diversifying and cross-compare the viewpoints of the various experts in	
committees and working groups (epidemiologists, primary healthcare providers, patients, etc.).	
4) Bringing in new heads of working groups and committees, new institutional	
representatives, etc., on a regular basis, so as to increase the number of	
experienced people and to enhance skills.	
5) Extending the requirement of transparency to all the work done by regulatory	
agencies and other competent authorities (including making available the	
documents used to develop positions or make decisions).	
6) Implementing a system of independent verification of declarations of	
interests.	
7) Implementing a system of sanctions in case of non-disclosure of interests.	
8) During meetings of committees or other working groups, hearing from the	
participants who have an interest in the company involved (either directly or	
as a competitor), e.g. the clinical trial investigators; then requiring all	
participants (experts or others) who have an interest (be it major or minor) in	
any company involved to leave the room, during the discussion leading up to a position being taken or a decision being made.	
9) Implementing and applying sanctions in case of participation of somebody in a	
position being taken or a decision made, in case of an interest in the company	
affected by the position or the decision.	
10) Maintaining a public register of all documents detained, as requested by the	
European ombudsman	
The mandate of EMA and national drug regulatory authorities	
The role of 'support to innovation', as understood by the EMA and national	
medicines agencies as optimising industry's return on investment frequently	
conflicts with the agencies' main mission of evaluating and regulating drugs and	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
medical devices. This <i>innovator</i> role, which also encompasses the provision of early scientific advice, should be closely monitored and subject to full transparency, so as to minimize regulatory capture. Those officials/experts participating in the provision of scientific advice should not be involved in the assessment of the pharmaceutical product at a later stage.	
The benefit-risk assessment of health products needs to be evidence-based	
Since 1965, the criteria for marketing authorisation in Europe are the demonstration of a medicine's efficacy, safety and quality. Efficacy is usually demonstrated in clinical trials, in which the effects of the new drug are often compared to placebo rather than to "the best therapeutic option available" for the same indication. And frequently surrogate outcomes are deemed sufficient which do not necessarily translate into tangible benefit for patients. Consequently, some drugs constitute a step backwards, unnecessarily exposing patients to adverse effects when other safer treatments exist.	
The paucity of new medicines that offer even a modest therapeutic advantage stands in stark contrast to the large number of new products that expose patients to unjustified risks. The majority of new medicines are "me-too" drugs and not "innovative" since they do not have an added therapeutic value (5,6).	
Rather than lowering the requirements for market authorisation of new drugs, as proposed in the adaptive pathways concept, the EMA should establish a compulsory demonstration of a new drug's therapeutic advance when compared to the best available therapeutic option. This would act as an incentive to reorient research and development towards unmet health needs -for which there's no adequate treatment -and true therapeutic progress.	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
Therapeutic advance should be the 4 th criterion to be demonstrated when applying for a marketing authorisation. The therapeutic advance of a new medicine would be appraised in comparison with existing treatments, and demonstrated by relevant clinical data collected from comparative clinical trials. Clinical trial results would then need to indicate the extent to which the new medicine would be more effective or safer than the existing standard treatment, specifying the relevant patient population.	
In order to implement more stringent criteria for market authorisation, concrete measures include: 1. The requirement for pharmaceutical companies filing marketing authorization applications to include complete results of clinical trials comparing the new drug against the drug(s) of reference, in their optimal conditions for use.	
2. A change in legislation at the European level requiring that marketing authorisation applications demonstrate the added therapeutic value and packaging safety of new drugs with a high level of evidence, demonstrated in the normal conditions of use.	
3. The provision of public financing for comparative clinical trials that allow drugs to be objectively rated among therapeutic strategies (including non-drug options), in terms of their risks and their benefits.	
Transparency in decision-making: access to documents is a right of EU Citizens and an institutional duty of the EMA	
The EMA's transparency requirements are enshrined in the EU directive 200/83/EC which regulates pharmaceutical products, as well as in the EU freedom of information Regulation (Regulation 1049/2001) which governs public access to documents at European Union's institutions and agencies.	
Health is a field where the decisions of EU institutions affect citizens' daily lives.	

Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
Public access to medicines safety and efficacy data contributes to informed decisions on treatment. Clinical trial data is not commercially confidential information	
Public access to full clinical data, including raw data, is particularly important to protect public health as it allows for comprehensive independent analysis, enhancing knowledge about the real effects of medicines and allowing comparative effectiveness reviews (9).	
Patients, consumers and healthcare professionals have long been deprived from having access to this important information. The EMA has prevented full disclosure under the guise that giving public access to commercially confidential information would jeopardize commercial interests.	
Implementation of the Clinical Trials Regulation	
The recently adopted EU Clinical Trials Regulation has the potential to significantly increase public access to clinical trial data. Regulators – and in particular the EMA in its key role of managing the EU clinical trials database- must uphold the principle that clinical trial data held by regulatory authorities is information of public interest. Any definition of commercially confidential information is to be interpreted narrowly and should never override disclosure of clinical trial data.	
The EMA must in fact fully comply with the Regulation on access to documents and the Treaty on the Functioning of the European Union, which identifies the "protection of health and life of humans" as an overriding public interest. (7) Moreover, under Regulation No 1049/2001 on access to documents, confidentiality is an exception: "In principle, all documents of the institutions should be accessible to the public. However, certain public and private interests should be protected by way of exceptions" (Regulation 1049/2001, recital 11).	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
In our view, the EMA's 'Draft proposal for an addendum, on transparency, to "the Functional Specifications for the EU portal and EU database to be audited" is unacceptable and at odds with the principles enshrined in Regulation (EU) No 536/2014 and the transparency advances it promised to bring, especially that of increasing the reliability and robustness of clinical data.	
The EMA's has the responsibility to protect and strengthen public health. However, its interpretation of Regulation (EU) No 536/2014 does little to meet the needs of patients and the public across the European Union yet goes a long way to soothe the requests of "clinical trial sponsors" such as the pharmaceutical industry, by introducing limited disclosure as the norm and by providing all the flexibilities sponsors need to circumvent their legal obligations to disclose clinical trial data. By allowing for redactions of clinical trial data, on the grounds of commercial confidentiality, the EMA is compromising public health and diminishing public trust in regulatory-decision making (10).	
A redefinition and narrowing of the notion of commercially confidential information is essential to prevent the EMA from relying solely on the self-classification by the sponsor of the information that may undermine the sponsor's economic interest or competitive position. Any exception to disclosure rules should only involve the removal of specific elements of information within a document and never be applied to an entire section or certain types of documents.	
Access to de-identified clinical trial participants' data: an essential step for secondary analysis	
EMA's views have dramatically changed since November 2012, when it announced that it would "proactively publish clinical- trial data and enable access to full data sets by interested parties" –to allow for reanalysis of trials' results.	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
It is important to distinguish patient personal data from de-identified participants' data. Participants accept to put themselves at risk, taking part in clinical trials, hoping that their participation will benefit society through the advancement of science. Furthermore, according to EU regulations, data submitted to regulatory authorities during a marketing authorisation procedure is submitted in non-identifiable form. Currently applied anonymisation methods safeguard patient confidentiality. Only in very specific cases (e.g., rare diseases) additional measures might be required to prevent re- identification.	
There is no public health rationale in preventing access to de-identified data by researchers and the European Medicines Agency should strive to ensure public access to these data in the future implementation of its access to clinical trials policies. Granting public access to raw data is crucial to minimise dangerous practices of reporting bias, which overrate the benefits of a drug while underestimating its harm.	
Moreover, industry-funded research often benefits from public funded research bodies (access to investigators and research teams at publicly research sites; public funding for basic research through EU grants and Member State funding, etc.). This is an additional argument that all data from biomedical research is made publicly available.	
Trade agreements should not hamper affordable access to needed medicines nor hinder clinical data transparency.	
As noted in the HMA/EMA draft strategy paper, political initiatives in the form of free trade agreements between the EU and non-EU countries increasingly include pharmaceuticals as an area of cooperation. This is the case in the Transatlantic Trade and Investment Partnership, the Comprehensive Economic and Trade Agreement and other free trade agreements with Japan, Singapore and South	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
Korea.	
The inclusion of a 'Pharmaceutical annex" with provisions on the regulation of pharmaceutical products in these trade agreements often undermines public health, for example, by including principles that should govern pricing and reimbursement decisions that limit the freedom of Member States to tailor their pricing and reimbursement strategies to provide sustainable access to medicines in favour of an increased voice for the pharmaceutical industry in these decision making processes (11). It is of utmost importance to ensure that agreements being currently negotiated: a) do not hamper by any means affordability of needed medicines, b) do not limit or restrain Member States' competence to negotiate price and reimbursement decisions c) do not impede public access to medicines' safety and efficacy data under the guise of trade secrets protection enshrined in trade agreements. Timely access to medicines shall not be in detriment of patient safety Whilst timely access to needed medicines is important, faster access should not take place to the detriment of patient safety. The concept of "innovative medicines" should be attributed to medicines addressing true unmet medical needs and with added therapeutic value when compared to the best available treatment.	
Existing flexibilities for market access - e.g. conditional approval, exceptional circumstances, compassionate use, accelerated assessment- should only be applied in duly justified circumstances. Communication to patients and their carers about the potential benefits of conditionally-approved medicines should not be overestimated, nor should their potential harms be underestimated. Treatment under conditional marketing authorisation must be closely monitored and any adverse drug reactions should be reported and published.	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
Adaptive pathways: deregulation under the guise of increased access, with patients and society picking up the tab	
According to data from the European Commission, the timelines for drug licensing have drastically shortened over the last 10-20 years, sometimes posing threats to patient safety (12). Premature licensing is achieved at the expense of proper evaluation, leading to more harm to patients (13, 14).	
Years of experience show that in Europe, the US and Canada, pharmaceutical companies frequently do not honour their commitments on post-authorisation evaluation of medicinal products (15,16,17).	
It should also be noted that the move to extend conditional marketing authorisation to all new medicines was rejected by the European Parliament and the Council in 2010. The current pharmacovigilance legislation further underscores that: "It is essential that a strengthened system of pharmacovigilance does not lead to the premature granting of marketing authorisations" (18).	
The EMA's 'adaptive pathways' approach – which builds on the proposal for an adaptive licensing approach to all new drugs (19) - raises numerous concerns from a public health point of view.	
First, adaptive pathways aims to grant marketing authorisations based on lower requirement for evidence, for instance by taking on board surrogate endpoints in detriment of clinically relevant outcomes to save costs and time. Since the marketing authorisation is granted based on limited data, patients will be potentially exposed to the harms of a medicine which has not been subject of a thorough evaluation. Evidence from the US from the last 16 years has shown that drugs approved once the legislation on expedited drug approvals had been passed were more likely to be withdrawn or receive a new black-box warning than drugs	

General comment (if any)	Outcome (if applicable)
	<to be="" by="" completed="" ema="" hma="" the=""></to>
authorised prior to the bill's passage (20).	
Second, there are potential consequences to patients' safety when the burden of evidence is shifted from pre-marketing to post-marketing. That also means that the risk is shifted to the patients and the cost to the public. The drug's evaluation is to be rolled out once the medicine is already on the market, but in reality post-authorisation commitments are often not honoured. It could prove extremely difficult to gather additional clinical data on a drug once it has been authorised.	
Third, the European Medicines Agency's pilot project, launched in March 2014, seems to be an ideal tool to circumvent democratic process. It paves the way for the deregulation of marketing approval procedures and increases industry's control over other healthcare stakeholders: health technology assessment (HTA) bodies (influence on pricing and reimbursement decisions), prescribers and patients (increased control over prescriptions, access to personal data, direct-to-consumer communication).	
Fourth, adaptive pathways come with an additional measure to the concept: "a prohibition on product liability suits during the initial marketing period" by injured patients or payers. This insidious measure clearly defends the interests of the manufacturers. Patients and healthcare professionals will not only have to agree to use a medicine which has not been adequately tested, but also end up not being able to prosecute the company if something goes wrong. This places desperate patients in a particularly vulnerable and unprotected position, which is clearly unethical.	
Fifth, the legal basis for a number of aspects of the adaptive pathways approach is missing, e.g. the power to force manufacturers to conduct post-licensing studies.	
Last but not least, the spill-over effect: Implementing adaptive pathways could lead to a situation whereby premature marketing authorisations become the rule	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
rather than the exception, even when no genuine public health need is identified, therefore putting EU citizens' health at risk.	
Scientific advice to pharmaceutical companies = risks of regulatory capture	
The provision of confidential "advice" to pharmaceutical companies on their development plans for new medicines in exchange for fees – is a potentially harmful practice that the EMA is now trying to extend to national health technology assessment (HTA) bodies in the European Union (EU) (21).	
The provision of scientific advice by regulators to the regulated, in exchange for fees, holds an inherent risk of regulatory capture. This is further accentuated when the committee responsible for providing advice on marketing authorisation procedures is concomitantly involved in scientific advice procedures.	
To minimise the risk of regulatory capture, committee members deciding on marketing authorisation should not be involved in the provision of scientific advice. Scientific advice should be transparent to allow independent scrutiny and enhance public trust. Detailed reports of the scientific advice provided by regulators to pharmaceutical companies during drug development and preregistration process should be published at the time of decision on trial, or not later than 12 months after the end of the trial. This information cannot be considered commercially confidential information as there is a clear overriding public interest in disclosure.	
Instead of providing customised advice to pharmaceutical companies, we urge the EMA to write up ad hoc guidelines that help drug manufacturers make development decisions that address genuine public health needs. Potential guideline deviation should be addressed through written exchange only and subject to transparency requirements (see above mentioned recommendations).	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
European public assessment reports (EPARs) and similar national regulatory documents should include an additional section summarising scientific advice given by the EMA at each stage of the development process. This information would not only facilitate better understanding of the data provided, but also allow for an assessment of the role of scientific advice in the approval of new medicines.	
Medicines agencies and price and reimbursement bodies shall collaborate while maintaining their different roles	
Cost-effectiveness assessment needs to remain independent from the Drug Regulatory Agencies. The EMA wants to be recognised as the "leading authority" in the evaluation and supervision of medicines. It intends to work more closely with health technology assessment (HTA) bodies to make sure that their assessments are not too divergent. It is necessary to recognise that the aims of EMA and HTA are not identical. Whereas for EMA efficacy, safety and quality are legally sufficient criteria, HTA needs to assess the comparative effectiveness measured in patient relevant outcomes (morbidity, mortality, quality of life). Pharmaceutical companies are increasingly challenging health technology bodies' recommendations when these do not serve their commercial interests. They would like HTA bodies to be bound by drug regulatory agency decisions.	
HTA bodies have expertise in comparing relative effectiveness of medicines as well as in cost-effectiveness assessment. They play a major role at the national level role in the sustainability of Member States' social insurance systems and should therefore remain fully independent of Drug Regulatory Agencies as well as from any influence of pharmaceutical companies.	
Rather than trying to "harmonise" the methods of HTA institutions and limit their	

by the EMA/HMA>

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
access to a listing of case summaries ("Narrative Case Summary").	<to be="" by="" completed="" ema="" hma="" the=""></to>
As major contributors of spontaneous reports of adverse drug reactions, it is unjustifiable for healthcare professionals, consumers and patients to have such limited access to EudraVigilance.	
In its draft revision document, the EMA proposed to share more data with marketing authorisation holders (MAH), which made sense since they are required to develop periodic benefit-risk evaluation reports about their drugs. Nevertheless, drug regulatory agencies have to closely monitor the MAH pharmacovigilance activities in order to avoid data being misinterpreted or withheld as recently happened on several occasions.	
The EMA also proposed to give research organisations, on request, "access to ICSR data sets similar to those provided for MAHs in response to justified research requests". However, the EMA set up restrictive conditions for granting access to researchers, e.g. the signature of confidentiality agreements. The EMA also demanded to "view any publication resulting from EudraVigilance data before submission (). [and that] any issues raised by the Agency () must be addressed to the satisfaction of the Agency before submission for publication". However, EMA's central role does not give it the right to monitor how the data are used or to censor scientific discussion.	
Anonymised narrative summaries of cases should be made available. Considerations about the re-identification of patient level data cannot be exaggerated. As rightly emphasised by EMA regulators "() standards for de-identifying personal data are available and continue to evolve to ensure adequate protection". ²² Additional safeguards can be applied in exceptional circumstances.	
We encourage the EMA in its Eudravigilance policy to support public health by: o proactively providing public access to useful qualitative data such as	

General comment (if any)	Outcome (if applicable)
anonymised summaries of cases; o granting public access to consumption data of drugs in the EU; o providing access to all drug regulatory authorities' assessment reports of MAH's periodic benefit-risk evaluation reports (former Periodic safety update reports); o not forcing researchers to sign "confidentiality agreements". PRAC Public Hearings: missing in action? Five years have passed since the adoption of the directive and regulation on Pharmacovigilance, another three years since the first PRAC meeting, but the PRAC Public Hearings have not yet been implemented. They are a long awaited and welcomed initiative but several major improvements are still needed to make the most of these hearings. We encourage the EMA to ensure that EU pharmacovigilance public hearings are as transparent and independent as the public sections of advisory committees in the USA (23).	<to be="" by="" completed="" ema="" hma="" the=""></to>
EMA's draft rules (published in 2014) allowed pharmaceutical companies to use public hearings as a platform to minimise/deny genuine safety concerns, as companies would be systematically granted "the opportunity to present its/their view(s) to the participants during the public hearing" by the EMA (1). In contrast, the US Food and drug administration (FDA) guidance on advisory committee prevents "the sponsor whose product is under review" from participating in the open panel of public hearings (2).	
The EMA proposed non-public hearings "where a marketing authorisation holder or another person intending to submit information that has confidential data relevant to the subject matter of the procedure" (1). We underline that non-public hearings hinder public scrutiny and should be reserved to protect whistleblowers, and should not offer MAHs an opportunity to influence the decision-making process.	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
Moreover, instead of being reluctant to organise live-broadcast and web- streaming of public hearings by adding everywhere the condition "when technically feasible", we expect the EMA to make the most of modern communication tools to ensure wider participation by the general public.	
PRAC 's role and independence should be reinforced The Pharmacovigilance Risk Assessment Committee (PRAC) is the committee at the European Medicines Agency that is responsible for assessing and monitoring safety issues for human medicines.	
On two recent occasions, the recommendations of the PRAC have not been duly followed:	
 On 10 January 2014, the PRAC recommended that Protelos/Osseor should no longer be used to treat osteoporosis, due to its risk of cardiovascular harm. Nonetheless, the CHMP opted not to recommend a suspension, but just introduced some restrictions to its use. 	
2) On 8 November 2013, the PRAC recommended the suspension of diacerein-containing medicines, due to their gastro-intestinal side effects and liver toxicity. Rather than accepting the PRAC position and withdraw the market authorisation(s), the Co-ordination Group for Mutual Recognition and Decentralised Procedures – Human just endorsed on 19 March 2014 a set of recommendations to restrict the use of diacerein- containing medicines.	
Both pharmaceutical products are still being marketed in the EU, despite their disproportionate risk of harm.	
Concrete measures to achieve a robust and proactive pharmacovigilance include:	

Gener	al comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
1)	Ensuring that decisions in pharmacovigilance matters are made independently from marketing authorisation committees.	
2)	Encouraging the undertaking and the public financing of post-marketing authorisation studies, as decided by the marketing authorisation or pharmacovigilance committees.	
3)	Applying sanctions, in particular financial penalties, for non-completion within the designated time period of post-marketing authorisation studies that marketing authorisation or pharmacovigilance committees have requested from marketing authorisation holders.	
4)	Publishing in a timely manner all pharmacovigilance data likely to encourage healthcare professionals and patients: to report the adverse effects experienced with this or that drug; to take special precautions; or to reconsider current treatments.	
5)	Making decisions to suspend or to withdraw marketing authorisation without delay, on the basis of an unfavourable risk-benefit balance, particularly when there is an alternative treatment with a better risk-benefit balance; with the benefit of the doubt given to the patient and not to the drug.	
6)	Requiring that the withdrawal of a drug from the market be preceded by online publication of the minutes of the pharmacovigilance committee that proposed the withdrawal, as well as the documents underlying that decision.	
Preve	nting otherwise avoidable medication errors	
The do	cument does not identify any priority in the prevention of medication	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
errors nor does it put forward measures to encourage the rational use of medicines. We urge the EMA to improve the quality of packaging to minimise the medication errors in practice. Guidelines on naming, labelling and packaging of medicinal products should be reviewed to proactively address patient safety concerns. In addition, packaging, labelling and package leaflets should be subject to user testing both in the hospital and in ambulatory settings. The comprehensive results of such tests should be thoroughly assessed by Drug Regulatory Agencies before granting a marketing authorisation.	
Encourage generic and biosimilar competition to enable affordable treatment	
Increasingly, and this is only exacerbated by the current economic crisis, Member States are under greater strain to provide universal access to care and to needed medicines. Most notably, more than 100 influential oncologists have described current prices of cancer medicines as: "astronomical, unsustainable and even immoral" (24). Recently, the exorbitantly high price of Sovaldi° (sofosbuvir) a new Hepatitis C drug was heavily criticised by NGOs, consumers, patients, carers, and healthcare professionals worldwide. Generic competition is an effective tool to bring medicine prices down. Prices tend to drop 25% a year after generic entry and 40% two years after entry (25,26). These savings translate across the health system: average savings are estimated to be almost 20 percent after the first year, and 25 percent after the second year (22). Unnecessary delays in generic and biosimilar market entry have a negative impact on drug affordability and increase the overall expenditure on medicines. Any decision to switch a prescription medicine to non-prescription must be evidence-based and have patient safety and rational use in mind	
The HMA/EMA proposed strategy paper refers to the need to ensure that mechanisms to re-classify medicines from prescription-only to non-prescription	

General comment (if any)	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
"are in place, effective and being used, thereby improving patient access"(1). We warrant extreme caution in any attempt to change the legal classification of medicines. These decisions should be evidence-based and have the best interests of patients in mind (i.e. high standards of patient safety). In addition, the potential for misuse and/or irrational use should be adequately weighed.	

Medical Devices: a priority not to be ignored

The EMA should also include another important aspect in its work plan to 2015: medical devices' evaluation. The medical devices market is rapidly expanding. The EMA should be structurally adjusted to be able to scientifically assess medical devices being marketed in the European Union. The US Food and Drug Administration can rely on the expertise of its Center for Devices and Radiological Health (CDRH) which is responsible for regulating companies that manufacture, repackage, re-label, and/or import medical devices sold in the United States. The EMA and the network of regulatory agencies, some of which are already responsible for regulating devices at national level, should further consider how to best address this important priority and uphold their responsibility to protect patients' health.

Specific comments on text

Line No. of the first line(s) affected	Comment and rationale; proposed changes	Outcome (if applicable) <to be="" by="" completed="" ema="" hma="" the=""></to>
Line 171 "Costly and complex" development of medicines	Debunking pharma myths on the costs of the current pharmaceutical model and its Research and Development The pharmaceutical industry generated higher profit margins than any other industrial sector in 2013, and is likely to have remained the most profitable sector in 2014. However, the majority of this revenue is not reinvested in R&D. The DG Competition enquiry revealed that between 2000 and 2007 pharmaceutical companies spent around 23% of their turnover on marketing and only 17% on R&D (23). The cost of a new drug discovery was claimed to be \$1.3bn	

(£834m; €1bn) in 2011, but this figure, which comes from the industry-supported Tufts Center, is likely to be at least a fourfold overestimation. Researchers have recalculated the Tufts Center figures using a more comprehensive methodology to include cheaper drugs in their calculation and drugs produced in part with public funds or tax credits. They found a mean cost closer to US\$90 million per new drug and a median cost of US\$60 million (27). Recent estimates from the same Tufts Centre have suggested that a new drug costs \$2.6 bn to develop (28). Consumer advocates and NGOs have criticized the new figures saying that critical information was missing from the analysis and that it was mere propaganda.

The real costs of R&D remain unknown – even the director of pharmaceutical company GlaxoSmithKline called the "1 billion estimate [of R&D costs] one of the greatest myths of the industry" - and estimates of the industry and independent analysts vary greatly.29,30

References:

¹ HMA, EMA. EU Medicines Agencies Network Strategy to 2020. Consultation draft, EMA/MB/151414/2015, March 27, 2015. Page 9.

² http://www.ema.europa.eu/ema/index.jsp?curl=pages/about_us/general_general_content_000130.jsp&mid=WC0b01ac0580029336

³ Food and Drug Administration. FY 2015: justification of estimates for appropriations committees.

www.fda.gov/downloads/aboutfda/reportsmanualsforms/reports/budgetreports/ucm388309.pdf.

⁴ Association Internationale de la Mutualité, International Society of Drug Bulletins, Medicines in Europe Forum, Nordic Cochrane Centre. "European Medicines Agency (EMA) softens its conflict of interest policy: Does this further open the door to undue influence instead of closing it?" Joint Press Release 28 November 2014. Available at: http://english.prescrire.org/en/79/549/49237/3990/3369/SubReportDetails.aspx

⁵ Prescrire Editorial Staff. "New drugs and indications in 2014" *Prescrire Int* 2015; **24** (159): 107-110.

⁶ Naci H et al. "Raising the bar for market authorisation of new drugs" *BMJ* 2012; **344**: e4261: 5 pages.

Official Journal of the European Union 31 December 2010: L 348/74-L 348/99.

⁷ European Ombudsman. Draft recommendation of the European Ombudsman in his inquiry into complaint 2560/2007/BEH against the European Medicines Agency, 2010

⁸ http://www.ombudsman.europa.eu/cases/decision.faces/en/11119/html.bookmark#hl10

⁹ Tucker M "How should clinical trial data be shared?" BMJ 2013; 347 doi: http://dx.doi.org/10.1136/bmj.f4465

¹⁰ Association Internationale de la Mutualité, Health Action International, International Society of Drug Bulletins, Medicines in Europe Forum "EU Clinical Trials Regulation: EMA steers away from transparency by misinterpreting exception provisions" Joint Press Release. February 2015. Available at: http://english.prescrire.org/en/79/549/49237/4104/4091/SubReportDetails.aspx

¹¹ https://www.oxfam.org/en/research/trading-away-access-medicines-revisited

¹² European Commission Directorate General Competition. "Final Report of the Pharmaceuticals Sector Enquiry"; 8 July 2009. ec.europa.eu :426 pages.

¹³ Carpentier D et al. "Drug review deadlines and safety problems" N Engl J Med 2008; **358**: 7 pages.

¹⁴ Frank C et al. "Era of faster FDA drug approval has also seen increased black-box warnings and market withdrawals" *Health Aff* 2014; **33**(8): 6 pages.

¹⁵ US Government Accountability Office "Drug safety – Improvement needed in FDA's postmarket decision-making and oversight process" Report GAO-06-402, 2006. www.gao.gov: 63 pages.

¹⁶ Carpentier D "Can expedited FDA drug approval without expedited follow-up be trusted" JAMA Internal Medicine 2014; **174** (1): 95-97.

¹⁷ Lexchin J "Notice of compliance with conditions: a policy in limbo" *Healthcare policy* 2007; **2** (4): 114-122 (+ annexes: 5 pages).

¹⁸ "Directive 2010/84/EU of the European Parliament and of the Council of 15 December 2010 amending, as regards pharmacovigilance, Directive 2001/83/EC on the Community code relating to medicinal products for human use"

¹⁹ Eichler H-G et coll. "Adaptive Licensing: Taking the Next Step in the Evolution of Drug Approval" Clinical Pharmacology & Therapeutics (2012); 91 (3): 426–437.

²⁰ - Frank C et al "Era of faster FDA Approval has also seen increased black-box warning and market withdrawals" *Health Affairs* 2014; 33(8): 1453-1459

²¹ AIM, HAI Europe, ISDB and MiEF "Providing "scientific advice" to pharma industry undermines the independence of regulatory authorities" Joint consultation response to the EMA's public consultation on its 'Best practice guidance for pilot EMA HTA parallel scientific advice procedures'; 15 July 2014: 6 pages. http://english.prescrire.org/en/79/549/49233/3696/3469/SubReportDetails.aspx

²² Eichler HG, Pétavy F,Pignatti F et al. Access to patient-level trial data- A boon to drug developers. NEJM doi:10.1056/NEJMp1310771 (2013)

²³ US Food and Drug Administration (FDA) "Guidance for the public, FDA advisory committee members, and FDA staff: the open public hearing at FDA advisory committee meetings" Final guidance; 15 May 2013: 13 pages.

²⁴ Laurance, J. (2013). Makers of anticancer drugs are "profiteering," say 100 specialists from around the world. *British Medical Journal (BMJ)* 346:f2810.

²⁵ Kanavos, P., Vandoros, S., Irwin, R., Nicod, E. and Casson, M. (2011). Differences in costs of and access to pharmaceutical products in the EU. *Policy department economic and scientific policy. Brussels.* 3.6. Access to generic medicines in EU member states and the benefits of generic competition: p70.

http://ec.europa.eu/competition/sectors/pharmaceuticals/inquiry/communication_en.pdf

²⁶ Executive Summary. Final Report DG Competition Sector Inquiry Summary.

²⁷ Light D and Lexchin J "Pharmaceutical Research and Development: What Do We Get for All that Money?" BMJ 2012; 344: 5 pages

²⁸ Mullard A "New drugs cost US\$2.6 billion to develop" *Nature Reviews Drug Discovery* 2014; **13**, 877.

²⁹ Light, D. and Lexchin, J. (2012). Pharmaceutical research and development: what do we get for all that money? *British Medical Journal (BMJ)*. 345.

Adams, B. (2013). GSK chief: drug prices should be lower. *PharmaTimes Online*. http://www.pharmatimes.com/article/13-03-18/GSK_chief_Drug_prices_should_be_lower.aspx [accessed August 6th, 2014].